Author’s response to reviews

Title: Knowledge, Attitudes and Practices of Cervical Cancer Prevention among Zambian Women and Men

Authors:

Anayawa Nyambe (anayawa@gmail.com)

Jarl Kampen (jarl.kampen@uantwerpen.be)

Stridutt Baboo (sridutt2002@yahoo.com)

Guido Van Hal (guido.vanhal@uantwerpen.be)

Version: 2 Date: 16 Apr 2019

Author’s response to reviews:

Dear Editor,

RE: Cover Letter for Addressing the Additional Editorial Request

Thank you for reviewing our manuscript. We hereby re-submit the edited version of our manuscript entitled, “Knowledge, Attitudes and Practices of Cervical Cancer Prevention among Zambian Women and Men”.

In accordance to the requirements of the Funding section, we have included information regarding the role of the funding body in design of the study and collection, analysis, and interpretation of data and in writing the manuscript. These changes are found on Page 27, line 640 to 642.

We appreciate your time and look forward to your response.

Sincerely,

Anayawa Nyambe (corresponding author)